Author's response to reviews

Title: Low-Grade Pelvic Masses with Spindle Cell and Fibroblastic Proliferation: Case Report

Authors:

John P Micha MD (bram@gynoncology.com)
Bram H Goldstein PhD (bramhg@hotmail.com)
Douglas Ellison MD (dellison@hoaghospital.org)
John V Brown III MD (jebbrown@cox.net)
Mark A Rettenmaier MD (mretten@cox.net)

Version: 3 Date: 20 April 2007

Author’s response to reviews: see over
April 20, 2007

Professor Michael Kidd
Editor-in-Chief
Journal of Medical Case Reports
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK
RE: 3533057891256349

Dear Professor Kidd:

Thank you for permitting us an opportunity to revise our manuscript entitled "A Rare Case of Low-Grade Pelvic Masses with Spindle Cell and Fibroblastic Proliferation". We appreciate the reviewers’ comments and have made every effort to try and accommodate their questions and concerns.

Please contact me if I can provide you with any additional information.

Sincerely yours,

John P. Micha, M.D.
Gynecologic Oncology Associates
351 Hospital Road, Suite 507
Newport Beach, CA 92663
(949) 642-5165
Reviewer #1

1) No revisions necessary. However, figure #1 is redundant and can be excluded.

We appreciate the reviewer’s suggestion. However, after conversing with the pathologist, he requested that we retain both figures.

Reviewer #2

1) Are there any reports that discuss recurrent disease and how was it handled?

Since this pathologic entity is so rare, there are no reported cases discussing the clinical management of this disease. In an attempt to better clarify the propensity for recurrent disease, we have since included an additional reference discussing the incidence of recurrence for this low-grade subtype (bottom of page 5).